Author’s response to reviews

Title: EXPLORING THE RELATIONSHIP BETWEEN THE BIG FIVE PERSONALITY CHARACTERISTICS AND DIETARY HABITS AMONG STUDENTS IN A GHANAIAN UNIVERSITY.

Authors
Freda Intiful (fdintiful@chs.edu.gh)
Emefa Oddam (emefatoddam@yahoo.com)
Irene Kretchy (ikretchy@ug.edu.gh)
Joana Quampah (quampah73@yahoo.com)

Version: 4 Date: 04 Feb 2019

Author’s response to reviews:

P.O. Box Kb 143
College of Health Sciences
SBAHS
University of Ghana
Korle-bu Accra
February 4th, 2019

The Editor
BMC Psychology

Dear Sir,

PSYO-D-17-00056R3
EXPLORING THE RELATIONSHIP BETWEEN THE BIG FIVE PERSONALITY CHARACTERISTICS AND DIETARY HABITS AMONG STUDENTS IN A GHANAIAN UNIVERSITY.

Once again thank you for your comments. The revisions made to the manuscript are as follows;

Declarations section (line 321)
  • Consent was clarified as written informed consent.

Figure legend (line 448)
  • Figure legend inserted.

Thank you for your consideration.

Yours sincerely,

Freda Intiful.

(Corresponding author)